Author's response to reviews

Title: Anti-invasive and antiangiogenic effects of MMI-166 on malignant glioma cells

Authors:

   Hiromichi Nakabayashi (blue_flare17@yahoo.co.jp)
   Toshio Yawata (tyawata117@yahoo.co.jp)
   Keiji Shimizu (kshimizu2017@yahoo.co.jp)

Version: 5 Date: 29 June 2010

Author's response to reviews:

Dear the editor of the BMC Cancer

In this revised version, we corrected only several expressions on a text. We have not changed the contents.

All authors have read and approved the manuscript.
Neither the entire paper nor any of its contents have been accepted by any other journal nor is the paper being submitted to any other journal.
We believe this paper may interest your readers. We hope that you will be able to look favorably on our request for publication in the BMC Cancer.

1. Title: Anti-invasive and antiangiogenic effects of MMI-166 on malignant glioma cells
2. Authors: Hiromichi Nakabayashi, M.D., Toshio Yawata, M.D., Keiji Shimizu, M.D.
3. Institutions: Department of Neurosurgery, Kochi University, Medical School, Nankoku, Japan,
4. Manuscript category: Research Article
5. Anatomical site: brain
6. Corresponding author: Hiromichi Nakabayashi, M.D.,
   Department of Neurosurgery, Kochi Medical School Kochi University, Kohasu, Oko-cho, Nankoku 783-8505, Japan
   Telephone number: (+81) 88-880-2397
   Fax: number: (+81) 88-880-2400
   e-mail address: blue_flare17@yahoo.co.jp

Sincerely yours,